Author's response to reviews

Title: Over-expressing Akt in T cells to resist tumor immunosuppression and increase anti-tumor activity

Authors:

Yanhong Wu (wuyanhong849@126.com)
Zhenling Deng (dengzhenling1985@126.com)
Yishu Tang (tangyishu111@163.com)
Shuren Zhang (zhangsr@cicams.ac.cn)
Yu-Qian Zhang (yqzhang.cams@gmail.com)

Version: 4
Date: 2 June 2015

Author's response to reviews:

Dear Editor,

The manuscript we submit has been revised according to your request:
1. Animal ethics statement with named IRB has been added (Line 103-105).
2. Acknowledgement section has been included (Line 355-361).
3. Competing Interest, Authors’ Contribution and Acknowledgement sections have been moved to before the Reference section (Line 347-361).
4. In the Authors’ Contribution section, we clarified that all authors read and approved the final manuscript (Line 353-354).

Thank you very much for helping us improve the quality of our manuscript.

Best Regards,

Yu-Qian Zhang, Ph.D.
Associate Professor of Department of Immunology
Cancer Institute, Chinese Academy of Medical Sciences
Beijing 100021, China
Email: yuqianzhang@hotmail.com